Author's response to reviews

Title: Ethanol extract of propolis protects macrophages from oxidized low density lipoprotein-induced apoptosis by inhibiting CD36 expression and endoplasmic reticulum stress-C/EBP homologous protein pathway

Authors:

Hua Tian (liutianfangyu@163.com)
Hongwei Sun (hongweisunyst@126.com)
Jiajun Zhang (gegewv_008@163.com)
Xiaowei Zhang (drgingfeng@163.com)
Li Zhao (zhaoli20131234@126.com)
Shoudong Guo (gsd20060501@163.com)
Yanyan Li (1228441486@qq.com)
Peng Jiao (jiaopeng196@163.com)
Hao Wang (tywanghao_2005@163.com)
Shucun Qin (shucunqin@hotmail.com)
Shutong Yao (yst228@126.com)

Version: 6 Date: 26 June 2015

Author's response to reviews: see over
Dear Editor:

Thank you very much for handling our manuscript entitled "Ethanol extract of propolis protects macrophages from oxidized low density lipoprotein-induced apoptosis by inhibiting CD36 expression and endoplasmic reticulum stress-C/EBP homologous protein pathway" (1651053095158727).

Editorial comments:

Please include details in your Methods section on the source of the propolis used to make the EEP used in your study.

The source of the propolis used to make EEP has been included in Methods section (Line 6, Page 6).

We would like to express our great appreciation to you for comments to our manuscript. We look forward to hearing from you.

Best regards.

Yours sincerely

Shutong Yao. (Corresponding author)
Institute of Atherosclerosis, Taishan Medical University
2# Yingsheng E Road, Taian, Shandong, 271000, P.R.China
Email: yst228@126.com